Author's response to reviews

Title: Physical exercise, fitness and dietary pattern and their relationship with circadian blood pressure pattern, augmentation index and endothelial dysfunction biological markers: EVIDENT study protocol

Authors:

Luis García-Ortiz (Lgarciao@usal.es)
Jose I Recio-Rodriguez (donrecio@gmail.com)
Carlos Martín-Cantera (Carlos.Martin@uab.es)
Alfredo Cabrejas-Sanchez (acabrejas@sescam.jccm.es)
Amparo Gomez-Arranz (amparoaga@hotmail.com)
Natividad Gonzalez-Viejo (nativgovejo@hotmail.com)
Eguskine Iturrgui-San Nicolas (EGUSQUINE.ITURREGISANNICOLAS@osakidetza.net)
Maria C Patino-Alonso (carpatino@usal.es)
Manuel A Gomez-Marcos (magomez@usal.es)
EVIDENT group (lgarciao@usal.es)

Version: 2 Date: 30 April 2010

Author's response to reviews: see over
COVER LETTER

Editor, BMC Public Health Journal
Salamanca, April 30, 2010

To the Editor of BMC Public Health Journal,

We submit the following manuscript reviewed:
Author list: Luis García-Ortíz, Jose I Recio-Rodriguez, Carlos Martín-Cantera, Alfredo Cabrejas-Sanchez, Amparo Gómez-Arranz, Natividad Gonzalez-Viejo, Eguskine Iturrgui-San Nicolas, Maria C Patino-Alonso and Manuel A Gómez-Marcos
Title: Physical exercise, fitness and dietary pattern and their relationship with circadian blood pressure pattern, augmentation index and endothelial dysfunction biological markers: EVIDENT study protocol
Journal: BMC Public Health
MS ID: 4669315923817667

Formatting changes to be made:
Major revisions (we require the author to make these changes)

Author groups: please include the name of the group within the list of authors, by adding the name to the ?first name? box; and the bracketed abbreviation to the ?last name? box in the submission system.

I have added Evident group in submission system

Please remove the names of the members of the group from the title page and place them in the Acknowledgements, as they are to be acknowledged as members of the study group.

I have removed Luis Garcia Ortiz, Manuel A Gómez Marcos and José I Recio Rodriguez and Carmen Patino Alonso from La Alamedilla health centre in the Acknowledgements because they are in Coordinator centre.
I’d like authors in title page same than in this paper: 

http://www.biomedcentral.com/1471-2458/10/33

The groups? contribution as a whole can be given in the authors? contributions section.

I have added in authors Contributions: The project will be developed by Evident group.

Trial registration number: Please make sure that this appears in the abstract in both the manuscript and submission system.

I have added in abstract in submission system: Trial Registration: Clinical Trials.gov Identifier: NCT01083082

Manuscript: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors' contributions; Authors? information (optional); Acknowledgements (optional); References

I have changed Methods/Design in abstract and manuscript

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Underlining: Please note that underlining is not supported in the final full text; you may wish to change it to bold.

I have removed the underlines

Authors' contributions: Please use initials to refer to authors' names.

I have changed the authors name by initial in Authors' contributions

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing
stage for your manuscript, and you will not be able to make any changes after acceptance.

I have changed y for and in authors name in title page and Authors' contributions

We look forward to your reply.

Yours faithfully,

Luis Garcia Ortiz.

Unidad de Investigación La Alamedilla.
Lgarcia@usal.es. Tel 0034 923 12 44 65